Author’s response to reviews

Title: Effectiveness of a Community Program for Older Adults with Type 2 Diabetes and Multimorbidity: A Pragmatic Randomized Controlled Trial

Authors:

John J. Miklavcic (miklavcic@chapman.edu)
Kimberly D. Fraser (kdfraser@ualberta.ca)
Jenny Ploeg (ploegj@mcmaster.ca)
Maureen Markle-Reid (mreid@mcmaster.ca)
Kathryn Fisher (fisheka@mcmaster.ca)
Amiram Gafni (gafni@mcmaster.ca)
Lauren E. Griffith (griffith@mcmaster.ca)
Sandra Hirst (shirst@ucalgary.ca)
Cheryl A. Sadowski (cherylas@ualberta.ca)
Lehana Thabane (thabanl@mcmaster.ca)
Jean A. C. Triscott (jean.triscott@ualberta.ca)
Ross Upshur (ross.upshur@utoronto.ca)

Version: 4 Date: 09 Apr 2020

Author’s response to reviews:

We have formatted the abstract as requested. We have added a statement related to written informed consent from all participants in the Ethics Approval and Consent to Participate section. We have included a statement on the availability of data. We have ensured that all figures and tables are cited in the text. We have uploaded the manuscript as a single, final, clean version without any tracked changes.